Author's response to reviews

Title: Clinics and Churches: Lifeworlds and health-seeking practices of older women with noncommunicable disease in rural South Africa

Authors:

Daniel Lopes Ibanez-Gonzalez (Daniel.IbanezLopes@gmail.com)
Stephen M Tollman (Stephen.Tollman@wits.ac.za)

Version: 5 Date: 5 May 2015

Author's response to reviews: see over
The editors

BMC International Health and Human Rights

Third rebuttal letter regarding additional comments on draft paper entitled “Clinics and Churches: Lifeworlds and health-seeking practices of older women with noncommunicable disease in rural South Africa”

We thank the editors for reviewing the revisions and hereby note that we have made the requested revision to the abstract by reducing its content to under 350 words and revising the subheadings.

If there is anything else which you require please let me know.

Yours Faithfully,

Dr Lopes Ibanez-Gonzalez
Developmental Pathways for Health Research Unit
Department of Paediatrics and Child Health
University of the Witwatersrand Medical School
7 York Road, Parktown, 2193
Johannesburg
South Africa

Email: Daniel.IbanezLopes@gmail.com